Author's response to reviews

Title: Prevalence and associated factors of anemia among non-pregnant women of reproductive age in Bangladesh: Multilevel logistic regression analysis

Authors:

Md Kamruzzaman (kamruzzaman@ru.ac.bd)
Md Golam Rabbani (mgrstat57@gmail.com)
Saw Aik (sawaik@hot.com)
Md Abu Sayem (sayem072003@yahoo.com)
Md Golam Hossain (hossain95@yahoo.com)

Version: 2
Date: 11 December 2014

Author's response to reviews: see over
To
Editor
BMC Public Health

Dear Editor
We submitted our manuscript of our paper “Prevalence and associated factors of anemia among non-pregnant women of reproductive age in Bangladesh: Multilevel logistic regression analysis” to your Journal. This is an original research work and it is a major public health problem in developing countries like Bangladesh. We revised according your suggestions. Secondary data that was collated by Bangladesh Demographic and Health Survey (BDHS) was used in our present study. BDHS described sampling technique, survey design, survey instruments, measuring system, quality control, ethical approval and subject consent in their report, so we only mentioned these terms in our methods. We hope our manuscript will be considered for publication. We are waiting for your positive response.

With best regards
Dr. Md. Golam Hossain
Professor of Health Research Group
Department of Statistics, University of Rajshahi
Rajshahi-6205, Bangladesh
E-mail: hossain95@yahoo.com
Tel. +880721750797 (Res.)